Author’s response to reviews

Title: Dental and Medical Health Status and Oral Health Knowledge among Visually Impaired and sighted Female Schoolchildren in Riyadh: A Comparative Study

Authors:

Salwa AlSadhan (Ssadhan@ksu.edu.sa)
Asma Al-Jobair (aaljober@ksu.edu.sa)
Mariam Bafaqeh (Mariam.sb@hotmail.com)
Hanadi Abusharifa (h.abusharifa@hotmail.com)
Maram Alagla (meemalagla@live.com)

Version: 2 Date: 13 Jun 2017

Author’s response to reviews:

POINT-BY-POINT RESPONSE LETTER

Editor Comments:

The authors should be reminded that the study comprised two cohorts of participants: visually impaired vs. sighted group, which should be reflected in the Title as well as the opening statement of the Discussion section. If that issue is addressed, I recommend that the manuscript should be considered for publication in BMC Oral Health.

- Adjusted as follows

Title:

Dental and Medical Health Status and Oral Health Knowledge among Visually Impaired and Sighted Female Schoolchildren in Riyadh: A Comparative Study

Discussion:

This comparative study provided information on the medical and dental health status, and the oral health knowledge of visually impaired and sighted female children aged 6–12 years living in Riyadh, Saudi Arabia.
Reviewer reports:

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English.

- This has been addressed in the previous revision as the whole manuscript was edited for English language through a professional English language editing service.

Declarations

- Ethics approval and consent to participate

A heading was added for it

- Consent to publish

Added as requested

All authors consent to the publication of this study

- Availability of data and materials

Present

- Competing interests

Present

- Funding

Added as requested

No external funds were used in this study

- Authors' Contributions

Present

- Acknowledgements

Present

- Authors' Information

Added as requested

- Salwa A. AlSadhan, BDS, MSc
Head of Community Service Unit
Associate Professor, Department of Periodontics and Community Dentistry,
College of Dentistry, King Saud University,
Riyadh, Kingdom of Saudi Arabia
Diriyah P. O. Box 60169, Riyadh 11545
Email: Ssadhan@ksu.edu.sa

- Asma M. Al-Jobair, BDS, MSc, Pedo Cert.
Associate Professor, Department of Pediatric Dentistry and Orthodontics,
College of Dentistry, King Saud University,
Riyadh, Kingdom of Saudi Arabia
Email: aaljober@ksu.edu.sa

- Mariam Bafaqeeh, BDS
Intern, College of Dentistry, King Saud University, Riyadh, Kingdom of Saudi Arabia
Email: Mariam.sb@hotmail.com

- Hanadi Abusharifa, BDS
Intern, College of Dentistry, King Saud University, Riyadh, Kingdom of Saudi Arabia
Email: h.abusharifa@hotmail.com

- Maram Alagla, BDS
Intern, College of Dentistry, King Saud University, Riyadh, Kingdom of Saudi Arabia
Email: meemalagla@live.com